\begin{table}%t1
\caption{\label{table.param}Parameters of the test simulations.}             
\par
%\centering
\small
\begin{tabular}{c c c c c c c c c c c c c }  
\hline\hline       
%&&&&&&&&&&&& \\[-8pt]
Application & $M~(M_{\odot})$ & $R~({\rm pc})$ & $T_{\rm n}~({\rm K})$ & $X$ & $Y$ & $\mu_{\rm n}$ & $\mu_{\rm i}$ & ${\dot {\cal N}}_{_{\rm LyC}}~({\rm s^{-1}})$ & $D~({\rm pc})$ & $R_{_{\rm St}}~({\rm pc})$ & Self-gravity & ${\cal N}_{_{\rm SPH}}$ \\
%&&&&&&&&&&&& \\[-8pt]
\hline                    
%&&&&&&&&&&&& \\[-8pt]
Sect. \ref{tempsmoothing} & 1000             & 1    & 10  &   0.7 & 0.3 & 2.35 & 0.678 &  $10^{49}$           &  0   &  0.189 & OFF & $10^6$\\
Sect. \ref{sec.hydro}     & 1000             & 1    & 10  &   0.7 & 0.3 & 2.35 & 0.678 &  $10^{49}$           &  0   &  0.189 & OFF & $3,6,10,20\times 10^5$\\  
Sect. \ref{sec:grav}      & $1.5\times 10^5$ & 14.6 & 10  &   0.7 & 0.3 & 2.35 & 0.678 &  $10^{49}$           &  0   &  1.43  & ON  & $10^7$\\
Sect. \ref{sec.rocket}    & 300              & 1    & 100 &   0.7 & 0.3 & 2.35 & 0.678 &  $10^{49}$           &  0.4 &  0.42  & OFF & $10^6$\\
Sect. \ref{sec.rdi}       & 20               & 0.5  & 100 &   1   & 0   & 1    & 0.5   &  $3.2\times 10^{48}$ &  3.5 &  --    & ON  & $3\times 10^5$\\
\hline                  
\end{tabular}
\end{table}